Author's response to reviews

Title: Clinical profile, outcomes and improvement in symptoms and productivity in rhinitic patients in Karachi, Pakistan

Authors:

Taimur Saleem (taimur@gmail.com)
Umair Khalid (uk2017@gmail.com)
Ubaid Sherwani (ubaid.sheerwani@gmail.com)
Shehzad Ghaffar (shehzadghaffar@aku.edu)

Version: 4 Date: 1 December 2009

Author's response to reviews:

Revisions made:

1. Full affiliations have been provided for the authors in the format required. Please note that no postal code was available for Ziauddin Medical college.

2. In the authors' contribution section, we have added the statement: "all authors read and approved the final manuscript".

3. In the abstract section, 'Background' has been used as a heading rather than 'Introduction'.

4. In the main manuscript, the 'Background' has been used instead of 'Introduction'.

5. Acknowledgements has been inserted directly after the authors' contributions.

6. Conflict of interest has been renamed as 'Competing interests'.

7. Tables has been modified so as to omit visible vertical lines.

8. All references have full list of authors (upto 30)

Thank you.